Optimal Fiscal Feedback on Debt in an Economy with Nominal Rigidities *

Tatiana Kirsanova[†] University of Exeter Simon Wren-Lewis[‡] University of Oxford

May 5, 2010

Abstract

We examine the impact of different degrees of fiscal feedback on debt in an economy with nominal rigidities where monetary policy is optimal. We look at the extent to which different degrees of fiscal feedback enhances or detracts from the ability of the monetary authorities to stabilise output and inflation. Using an objective function derived from utility, we find the optimal level of fiscal feedback to be small. There is a clear discontinuity in the behaviour of monetary policy and welfare either side of this optimal level. As the extent of fiscal feedback increases, optimal monetary policy becomes less active because fiscal feedback tends to deflate inflationary shocks. However this fiscal stabilisation is less efficient than monetary policy, and so welfare declines. In contrast, if fiscal feedback falls below some critical value optimal monetary policy becomes strongly passive, and this passive monetary policy leads to a sharp deterioration in welfare.

Key Words: Fiscal Policy, Feedback Rules, Debt, Macroeconomic Stabilisation JEL Reference Number: E52, E61, E62, E63

^{*}We would like to thank Richard Dennis, John Driffill, Martin Ellison, Campbell Leith, Paul Levine, Patrick Minford, Joe Perlman, Alan Sutherland, and participants at a conference organised by the Federal Reserve Bank of Atlanta (particularly Eric Leeper, James Nason, Stephanie Schmitt-Grohe, Chris Sims, and Martin Uribe), as well as referees and the editor, for helpful comments and discussions. Any errors are ours.

[†]Corresponding author. Address: University of Exeter, School of Business and Economics, Streatham Court, Rennes Drive Exeter EX4 4PU; e-mail: t.kirsanova@exeter.ac.uk

[‡]Address: University of Oxford; e-mail: s.wren-lewis@economics.ox.ac.uk

1 Introduction

With the occasional and notable exception, most governments now see one of their primary economic responsibilities as ensuring that the national debt stays within reasonable bounds. In some cases explicit targets for the debt to GDP ratio have been announced, with the implication that if debt deviates from this target, some form of 'fiscal feedback' via taxes or spending will operate. However it is also recognized that any attempt to control the debt stock, or the public sector deficit, too tightly may induce instability in other macroeconomic variables. In this paper we examine this trade-off.

We work with a standard New Keynesian model of a closed economy with Calvo pricing to examine how optimal monetary policy varies with the degree to which fiscal instruments respond to the state of government indebtedness, as measured by fiscal feedback. Traditionally it was thought that some minimum level of fiscal feedback was required for a stable and determinate economy. However, the Fiscal Theory of the Price Level literature¹ has argued that a determinate equilibrium may be possible when no feedback occurs, but where prices adjust to ensure the government's intertemporal budget constraint holds. If fiscal feedback is strong, such that fiscal adjustments ensure the government's budget is satisfied, monetary policy is 'active' in the sense that it focuses on the control of inflation; if the fiscal feedback is negligible, monetary policy becomes 'passive' in that it must also ensure fiscal solvency. This paper provides a systematic treatment of the nature of equilibrium over the range of fiscal feedback parameters.²

Our analysis forms a bridge between the literature on optimal monetary policy and the literature on joint monetary-fiscal optimization. The former work, as exemplified by Woodford (2003), examines optimal monetary responses to shocks where there are lump-sum taxes available to continuously satisfy the government budget constraint. The latter literature, and in particular Benigno and Woodford (2004), Schmitt-Grohe and Uribe (2004) and Schmitt-Grohe and Uribe (2007), have looked at joint optimization where there are no constraints on the movement of at least one fiscal instrument. However, neither of these approaches allow us to examine the implications that the (mis)conduct of fiscal policy can have on the operation of optimal monetary policy. Here we do this by computing optimal monetary policy *conditional* on fiscal actions in the form of simple debt-controlling rules. This description of fiscal policy appears to be a more realistic modelling of current institutional arrangements. It is generally the case that fiscal policy

¹See Leeper (1991), Woodford (1996), but also Buiter (2002) for a more critical view.

²The terms 'active' and 'passive' monetary policy are derived from Leeper (1991). Leeper (1991) also describes strong fiscal feedback as 'passive fiscal policy' and negligable fiscal feedback as 'active fiscal policy'.

is far less flexible than monetary policy, and partly as a result, the focus of policy makers seems to be on how quickly government indebtedness should be corrected, as the debates around the Stability and Growth Pact of the European Monetary Union illustrate.³

We provide a welfare ranking of policies based on a measure derived from consumer utility, and we also compute joint monetary-fiscal optimization as a benchmark. We find that active monetary policy combined with sufficient fiscal feedback clearly dominates a passive monetary policy when fiscal feedback is negligible. The optimal degree of fiscal feedback occurs when debt follows a path very close to a unit root process, mirroring the path of debt under joint monetaryfiscal optimization. When government spending is the fiscal instrument, this optimal degree of fiscal feedback produces a level of welfare that is only very slightly worse that joint monetaryfiscal optimization, suggesting that the costs of restricting fiscal policy to a simple debt feedback rule are negligible in this case.

However, we find that there are two cases where optimal monetary policy can take on passive features (in the sense that interest rates initially fall following a cost-push shock) even when there is significant fiscal feedback. The first is when the initial level of debt is high, so that changes in interest rates are a particularly effective way of managing debt. The second is where the fiscal feedback parameter is too large from a welfare perspective, such that this strong feedback stabilizes inflation as well as debt. As a result, monetary policy is no longer required to play an active stabilization role, but this substitution is inefficient in terms of social welfare.

We find that the strength of fiscal feedback affects the stability properties of the economy. When fiscal feedback is either completely absent or above a certain threshold optimal monetary policy is able to stabilize both inflation and debt. When fiscal feedback is non-zero but small, we find that optimal monetary policy chooses policies that allow debt to follow a mildly explosive path. This path for debt implies that the economy as a whole does not return to a stationary equilibrium after a temporary shock. Nevertheless, discounting implies that this mildly explosive behavior represents the optimal outcome for monetary policy.

Our initial results are derived assuming that government spending is the fiscal instrument in the feedback rule. We then examine a feedback rule that uses income taxes as an instrument. This produces similar results, although in this case feedback that is too strong (in welfare terms) has less of an impact on monetary policy and welfare. We also examine the robustness of our results to replacing 'infinitely lived' consumers with consumers who face a constant probability of

³This does not imply fiscal policy makers are 'irrational', but may simply reflect overriding political economy concerns that are outside the scope of this paper.

death (and leave no bequests). In this case, joint monetary-fiscal optimization no longer implies a unit root process for debt. While other elements of our results remain unchanged, this analysis does highlight the intergenerational aspects of different degrees of fiscal feedback.

The paper is organized as follows. Section 2 outlines our core model, where consumers are 'infinitely lived'. In Section 3 we present the case of joint monetary-fiscal optimization, which provides a benchmark for our analysis. Section 4 presents our main results, where we use a feedback rule from debt to government spending. We discuss the method and form of the solution in Section 4.2, and then examine its properties in terms of monetary policy and welfare in Section 4.3. Section 5 examines the robustness of our results. We first consider using taxes, or taxes and spending, as fiscal instruments (Section 5.1), then vary the level of steady state debt (Section 5.2) and consider different shocks (Section 5.3). Finally we adapt our model to include consumers with finite lives, and look at intergenerational issues (Section 5.4). Section 6 concludes.

2 The Model

2.1 Consumers

Our model of the household sector is familiar from Woodford (2003). Our economy is inhabited by a large number of individuals, who specialize in the production of a differentiated good (indexed by z), and who spend h(z) of effort in its production. They consume a basket of goods C, and derive utility from per capita government consumption G. Individuals' maximization problem is

$$\max_{\{C_v, h_v\}_{v=t}^{\infty}} \mathcal{E}_t \sum_{v=t}^{\infty} \beta^{v-t} \left[u(C_v, \xi_v) + f(G_v, \xi_v) - v(h_v(z), \xi_v) \right].$$
(1)

Here ξ is a preference shock. The price of a differentiated good z is denoted by p(z), and the aggregate price level is P. An individual chooses optimal consumption and work effort to maximize criterion (1) subject to the demand system and the flow budget constraint:

$$P_t C_t + \mathcal{E}_t \left(Q_{t,t+1} \mathcal{A}_{t+1} \right) \le \mathcal{A}_t + (1 - \Upsilon_t) \left(w_t(z) h_t(z) + \Pi_t(z) \right) + T, \tag{2}$$

where $P_t C_t = \int_0^1 p(z)c(z)dz$ is nominal consumption, \mathcal{A}_t are nominal financial assets of a household, Π_t is profit and T is a constant lump-sum tax/subsidy. Here w is the wage rate, and Υ_t is a tax rate on income. $Q_{t,t+1}$ is the stochastic discount factor which determines the price in period t to the individual of being able to carry a state-contingent amount \mathcal{A}_{t+1} of wealth into period t + 1. The riskless short term nominal interest rate i_t has the following representation in terms of the stochastic discount factor:

$$\mathcal{E}_t(Q_{t,t+1}) = \frac{1}{(1+i_t)}.$$

Each individual consumes the same basket of goods. Goods are aggregated into a Dixit and Stiglitz (1977) consumption index with the elasticity of substitution between any pair of goods given by $\epsilon_t > 1$ (which is a stochastic elasticity with mean ϵ^4), $C_t = \left[\int_0^1 c_t^{\frac{\epsilon_t - 1}{\epsilon_t}}(z) dz\right]^{\frac{\epsilon_t}{\epsilon_t - 1}}$. We assume that the not present value of the val

We assume that the net present value of individual's future income is bounded. We also assume that the nominal interest rate is positive at all times. These assumptions rule out infinite consumption and allow us to replace the infinite sequence of flow budget constraints of the individual by a single intertemporal constraint,

$$\mathcal{E}_t \sum_{v=t}^{\infty} Q_{t,v} C_v P_v \le \mathcal{A}_t + \mathcal{E}_t \sum_{v=t}^{\infty} Q_{t,v} \left\{ (1 - \Upsilon_v) \left(w_v(z) h_v(z) + \Pi_v(z) \right) \right\} + T \right\}.$$
(3)

The optimization requires that the household exhaust its intertemporal budget constraint and, in addition, the household's wealth accumulation must satisfy the no Ponzi game condition:

$$\lim_{s \to \infty} \mathcal{E}_t \left(Q_{t,s} \mathcal{A}_s \right) = 0. \tag{4}$$

We assume the specific functional form for the utility from consumption component, $u(C_v, \xi_t) = \frac{(C_v \xi_t)^{1-1/\sigma}}{1-1/\sigma}$. Household optimization leads to the following dynamic relationship for aggregate consumption:

$$C_t = \mathcal{E}_t \left(\left(\frac{1}{\beta} \frac{P_{t+1}}{P_t} Q_{t,t+1} \right)^{\sigma} C_{t+1} \frac{\xi_{t+1}}{\xi_t} \right).$$
(5)

Additionally, aggregate (nominal) asset accumulation is given by

$$\mathcal{A}_{t+1} = (1+i_t) \left(\mathcal{A}_t + (1-\Upsilon_t) \left(W_t N_t + \Pi_t \right) - P_t C_t - T \right), \tag{6}$$

where W_t and N_t are aggregate wages and employment.

We linearize equation (5) around the steady state (here and everywhere below for each variable X_t with steady state value X, we use the notation $\hat{X}_t = \ln(X_t/X)$). Equation (5) leads to the following Euler equation (intertemporal IS curve):

$$\hat{C}_{t} = \mathcal{E}_{t}\hat{C}_{t+1} - \sigma\left(\hat{\imath}_{t} - \mathcal{E}_{t}\hat{\pi}_{t+1}\right) + \xi_{t+1} - \xi_{t}.$$
(7)

Inflation is $\pi_t = \frac{P_t}{P_{t-1}} - 1$ and we assume steady state inflation is zero.

⁴We make this parameter stochastic to allow us to generate shocks to the mark-up of firms.

2.2 Price Setting

Price setting is based on Calvo contracting as set out in Woodford (2003). Each period agents recalculate their prices with fixed probability $1-\gamma$. If prices are not recalculated (with probability γ), they remain fixed. Following Woodford (2003) and allowing for government consumption terms in the utility function, we can derive the following Phillips curve for our economy⁵:

$$\hat{\pi}_t = \beta \mathcal{E}_t \hat{\pi}_{t+1} + \frac{(1 - \gamma \beta)(1 - \gamma)\psi}{\gamma \left(\psi + \epsilon\right)} \hat{s}_t,\tag{8}$$

where marginal cost is

$$\hat{s}_t = \frac{1}{\psi}\hat{Y}_t + \frac{1}{\sigma}\hat{C}_t + \frac{\tau}{(1-\tau)}\hat{\Upsilon}_t - \hat{\zeta}_t + \left(\frac{v_{y\xi}}{v_y} - \frac{u_{C\xi}}{u_C}\right)\hat{\xi}_t + \hat{\eta}_t.$$

The shock $\hat{\eta}_t$ is a mark-up shock and $\hat{\zeta}_t$ is a technology shock, as we assume the production function $y_t = Z_t h_t$, $Z_t = Z_{t-1}\zeta_t$, where ζ_t has a mean of unity. Here $\psi = v_y/v_{yy}y$ and τ is the steady state income tax rate.

Under flexible prices and in the steady state the real wage is always equal to the monopolistic mark-up $\mu_t = -(1 - \epsilon_t)/\epsilon_t$. Optimization by consumers then implies:

$$\frac{\mu^w}{\mu_t} = \frac{v_y(y_t^n(z), \xi_t)}{\left(1 - \hat{\Upsilon}_t^n\right) u_C(C_t^n, \xi_t)},\tag{9}$$

where superscript n denotes natural levels (see Woodford (2003)), and μ^w is a steady state employment subsidy which we discuss below. Linearization of (9) yields

$$\hat{Y}_{t}^{n}\frac{1}{\psi} + \hat{C}_{t}^{n}\frac{1}{\sigma} + \frac{\tau}{(1-\tau)}\hat{\Upsilon}_{t}^{n} - \hat{\zeta}_{t} + \left(\frac{v_{y\xi}}{v_{y}} - \frac{u_{C\xi}}{u_{C}}\right)\hat{\xi}_{t} = 0.$$
(10)

2.3 Fiscal Constraint

The government buys goods (G_t) , taxes income (with tax rate Υ_t), raises lump-sum taxes, pays an employment subsidy and issues nominal debt \mathcal{B}_t . The evolution of the nominal debt stock can be written as:

$$\mathcal{B}_{t+1} = (1+i_t)(\mathcal{B}_t + P_t G_t - \Upsilon_t P_t Y_t - T + \mu^w).$$

The employment subsidy (μ^w) and lump-sum taxes (T) are constant and cannot be used to stabilize the economy. This equation can be linearized as (defining $B_t = \mathcal{B}_t/P_{t-1}$ and denoting

 $^{{}^{5}}$ The derivation is identical to the one in Woodford (2003), amended by the introduction of mark-up shocks as in Beetsma and Jensen (2004).

the steady state ratio of debt to output as χ):

$$\chi \hat{B}_{t+1} = \chi \hat{i}_t + \frac{1}{\beta} \left(\chi \hat{B}_t - \chi \hat{\pi}_t + (1 - \rho) \, \hat{G}_t - \tau \left(\hat{\Upsilon}_t + \hat{Y}_t \right) \right). \tag{11}$$

where $\rho = C/Y$ in steady state.

2.4 Aggregate Relationships

Output is distributed as wages and profits:

$$Y_t = W_t N_t + \Pi_t. \tag{12}$$

Government expenditures constitute part of demand, so the national income identity can be written as

$$Y_t = C_t + G_t, (13)$$

and in steady state $G = (1 - \rho)Y$. The linearized national income identity is then:

$$\hat{Y}_t = (1 - \rho)\hat{G}_t + \rho\hat{C}_t.$$
 (14)

2.5 Behavior of the Economy

We now write down the final system of equations for the 'law of motion' of the out-of-steady-state economy. We simplify notation by using lower case letters to denote 'gap' variables, where the gap is the difference between actual levels and natural levels i.e. $x_t = \hat{X}_t - \hat{X}_t^n$. The model consists of an intertemporal IS curve (15), the Phillips curve (16), national income identity (17), and an equation explaining the evolution of debt (18). We could use the linearized assets accumulation equation (6) instead of (18) as they are equivalent (equation (19)). The system is:

$$c_t = \mathcal{E}_t c_{t+1} - \sigma(i_t - \mathcal{E}_t \pi_{t+1}), \tag{15}$$

$$\pi_t = \beta \mathcal{E}_t \pi_{t+1} + \kappa \left(\frac{1}{\sigma} c_t + \frac{1}{\psi} y_t + \frac{\tau}{(1-\tau)} \tau_t + \hat{\eta}_t \right), \tag{16}$$

$$y_t = (1 - \rho)g_t + \rho c_t, \tag{17}$$

$$\tilde{b}_{t+1} = \chi i_t + \frac{1}{\beta} \left(\tilde{b}_t - \chi \pi_t + (1 - \rho) g_t - \tau (\tau_t + y_t) \right) + \hat{\delta}_t,$$
(18)

$$\tilde{a}_t = \tilde{b}_t,\tag{19}$$

where parameter $\kappa = \frac{(1-\gamma\beta)(1-\gamma)\psi}{\gamma(\psi+\epsilon)}$ and $\hat{\delta}_t = \frac{\chi(1-\beta)}{\beta}\hat{\xi}_t - \left(\frac{(1-\beta)}{\beta} + \frac{(1-\rho_q)}{\sigma}\right)\frac{\chi\psi\sigma}{(\sigma+\psi)}\hat{\zeta}_t$ is a composite shock. We denote $\tilde{b}_t = \chi \hat{B}_t$. Note that preference and technology shocks only appear in so far as they impact on debt, while cost push shocks matter through the Phillips curve.

It remains to specify policy. We do this in Section 3, where we also discuss some benchmark results of full optimization.

2.6 Calibration

We take the model's frequency to be quarterly. To achieve a steady state rate of interest of approximately 4%, we set the household discount rate β to 0.99. Output is normalized to one, and the ratio of government consumption to output, $1 - \rho$, is 0.25, which determines the relative preference for government spending in utility. The remaining parameters of the utility function are typical of those used in the literature, see e.g. Canzoneri et al. (2006). The elasticity of intertemporal substitution σ is taken as 1/1.5, the Calvo parameter γ is set at 0.75 so as to imply average contracts of about a year, the elasticity of demand is taken as $\varepsilon = 7.0$ to achieve a 17% mark up, and the elasticity of labour demand is taken as $\psi = 1/3$.

We consider three values for the debt to GDP ratio. Our 'high' debt level corresponds to 60% of annual output, which is the level of debt in a number of European economies. For analytical purposes, it is useful to consider a 'low debt' case where the steady state debt is zero. (Although unusual, such cases are not unknown: New Zealand, Sweden and Ireland have net debt to GDP ratios close to zero.) Our base case for χ is the midpoint of 30% of annual output. We discuss these figures further in Section 5.2 below.

Following Ireland (2004) and Canzoneri et al. (2006), the preference shock is calibrated as an AR(1) process $\hat{\xi}_t = \rho_{\xi}\hat{\xi}_{t-1} + \varepsilon_{\xi t}$ with $\rho_{\xi} = 0.9$ and $\sigma(\varepsilon_{\xi t}) = 0.03$. We calibrate the productivity shock as an AR(1) process $\hat{\zeta}_t = \rho_{\zeta}\hat{\zeta}_{t-1} + \varepsilon_{qt}$ with $\rho_{\zeta} = 0.9$ and $\sigma(\varepsilon_{\zeta t}) = 0.0075$. This is broadly in line with the values used in Canzoneri et al. (2006) $(\rho_{\zeta}, \sigma(\varepsilon_{\zeta t})) = (0.92, 0.0090)$, Ireland (2004) $(\rho_{\zeta}, \sigma(\varepsilon_{\zeta t})) = (1.00, 0.0109)$ and those used in Schmitt-Grohe and Uribe (2007) $(\rho_{\zeta}, \sigma(\varepsilon_{\zeta t})) = (0.86, 0.0064)$. ⁶ Among these three studies, only Ireland (2004) uses a cost-push shock, which is AR(1) with a standard deviation of 0.0044. Smets and Wouters (2003) reports an i.i.d. cost push shock with a much smaller standard deviation in the model with inflation persistence, while Rudebusch (2002) estimates a standard deviation of 0.01 for an i.i.d. cost push shock. In the analysis below, we calibrate the standard deviation for inflation of 0.0038 that is the same order of magnitude as empirical data in developed countries over the last couple of decades.

⁶Schmitt-Grohe and Uribe (2007) do not consider either taste shocks or cost-push shocks.

3 Joint Optimization as a Benchmark Case

The primary aim of this paper is to study the effect of a fiscal policy on monetary policy decisions. As a benchmark case we first compute a fully optimal policy i.e. joint monetary-fiscal optimization. We assume that both authorities jointly set their instruments $\{i_t, g_t, \tau_t\}$ to maximize the aggregate utility function:

$$\max_{\{i_s\}_{s=t}^{\infty}} \frac{1}{2} \mathcal{E}_t \sum_{s=t}^{\infty} \beta^{s-t} \left[u(C_s) + f(G_s) - \int_0^1 v(h_s(z)) dz \right].$$
(20)

We show in the working paper version of this paper Kirsanova and Wren-Lewis (2007), denoted hereafter as WP^7 that problem (20) implies the following optimization problem:

$$\min_{\{i_s\}_{s=t}^{\infty}} \frac{1}{2} \mathcal{E}_t \sum_{s=t}^{\infty} \beta^{s-t} \left[a_\pi \pi_s^2 + a_c c_s^2 + a_g g_s^2 + a_y y_s^2 \right] + \mathcal{O}(3),$$
(21)

where $\mathcal{O}(3)$ collects terms of higher than second order and terms independent of policy, and all a_i are positive. This quadratic approximation to social welfare is obtained assuming that there is a constant production subsidy $\mu^w = T$ that eliminates the distortion caused by monopolistic competition and income taxes in steady state. (This approach follows Woodford (2003). Sutherland (2002) and Benigno and Woodford (2004) use an alternative way of eliminating first order terms from welfare, while Schmitt-Grohe and Uribe (2004) do not use a linear quadratic framework, but instead adopt a Ramsey approach.) The steady state government spending is chosen optimally as is standard in the literature. We assume that the authorities have sufficient credibility to commit to the time inconsistent plan, so it can implement the first best time inconsistent solution.

Note that expression (21) contains a quadratic term in government spending, g. This term enters the welfare expression because it is assumed in (1) that households derive utility from the consumption of public goods, and that the level of government spending in steady state reflects this. However, if we instead assumed that government spending was pure waste, but the government still used g as a policy instrument, changes in g would still influence social welfare through the national income identity, but it would not constitute an independent source of welfare loss.

In order to solve for the fully optimal policy we specify system matrices and use MATLAB code by Söderlind (1999). The procedure is straightforward and non-innovative, so we only discuss the results, which are summarized in Table 1.

⁷It is available from www.people.ex.ac.uk/tkirsano and upon request from authors.

Shocks	Welfare loss of fully optimal monetary and		
applied	fiscal policy if fiscal instruments are:		
	spending	taxes	spending and taxes
$\operatorname{cost-push}$	2.054	0.230	0.220
taste	2.53×10^{-5}	1.25×10^{-6}	1.19×10^{-6}
productivity	4.49×10^{-5}	2.22×10^{-6}	2.12×10^{-6}
all shocks	2.054	0.230	0.220

Table 1: Welfare implications of shocks, measured as percent of steady state consumption under monetary-fiscal optimisation.

Table 1 shows that cost-push shocks have a much greater impact on social welfare than preference or productivity shocks when policy is optimal. This is a well known result, and arises because mark up shocks change the relationship between inflation and output, both of which are key policy objectives. In contrast, the impact of preference or productivity shocks only matter through their impact on debt: if lump sum taxes were available as a stabilization policy instrument, monetary policy could fully offset these shocks. Although lump sum taxes are constant in our case, the behavior of debt discussed below means that shocks to debt have little impact on welfare, see also Leith and Wren-Lewis (2007). Our analysis below is based on applying the full menu of shocks, but as cost-push shocks are likely to dominate we also consider a case where there are only preference and productivity shocks (Section 5.3).

Figure 1 shows the path of key variables following a unit cost-push shock. The most straightforward case is where only government spending is available as an instrument. Here monetary policy responds to the increase in inflation generated by the cost-push shock, raising interest rates in both the initial and subsequent periods. Higher interest rates raise the level of debt, which increases gradually but eventually stabilizes at a new higher level. This unit root process for steady state debt, which is noted for joint monetary-fiscal optimization in Benigno and Woodford (2004) and Schmitt-Grohe and Uribe (2004), is an extension of a result from the tax smoothing literature (Barro (1979)). As the welfare function is convex and there is discounting, it is preferable to move fiscal instruments by a small amount permanently to service a new higher level of debt, rather than change them by a large amount on a temporary basis to return debt to its initial level. Note, however, that the cut in government spending is larger in the first period of the simulation than subsequent periods, so there is a very modest attempt to reduce the size of the long run increase in debt. The reasons for this, and its implications for policy under discretion, are discussed in Leith and Wren-Lewis (2007). When income taxes are available as a fiscal instrument, then there is an initial attempt to directly offset to cost-push shock by cutting taxes. Lower income taxes increase the incentive to work, which directly reduces the inflationary consequences of the cost-push shock.

4 Fiscal stabilization of debt

4.1 Simple Feedback Rules

A number of authors have used simple fiscal rules where a fiscal instrument responds to deviations in debt from some reference value (for example Schmitt-Grohe and Uribe (2007), Canzoneri et al. (2006) among others). In the first part of our analysis we focus on the use of government spending as an instrument, but we broaden it to include income taxes subsequently. Empirical estimates of fiscal policy reaction functions have tended to focus on cyclical behavior rather than debt feedback (see Favero and Monacelli (2005), Taylor (2000), Auerbach (2002) for example), but the evidence suggests that both government spending and taxes do move to stabilize debt (see Muscatelli et al. (2004) for example). Our initial focus on spending is expositionally convenient, for reasons that become clear when we consider taxes in Section 5.1. We postulate that out-ofsteady-state government expenditure G_t is related to out-of-steady-state debt according to the following simple feedback rule:

$$G_t - G_t^n = -\lambda \left(B_t - B \right), \tag{22}$$

Note that this rule relates to gap variables, which implies that actual government spending will adjust with changes in the natural level of this variable. The latter assumption follows naturally from the assumption that steady state government spending is chosen optimally, and is standard in the literature on optimal policy.

Log-linearization of this rules yields

$$g_t = -\frac{\lambda}{(1-\rho)}\tilde{b}_t,\tag{23}$$

Simple mechanistic rules for fiscal policy more accurately reflect institutional rigidities in fiscal policymaking than full optimization, where the latter would imply that fiscal instruments would immediately respond in an optimal fashion to contemporaneous shocks. Of course there are a variety of potential simple rules, but as our focus in this paper is on how debt stabilization affects optimal monetary policy, the specification above seems appropriate. In addition, we note that debt is the only state variable in our model. Furthermore, as we show below, when government

spending is the instrument this rule comes very close to reproducing the outcome that would occur under full optimization, and so more complicated rules appear unnecessary in this case.

In what follows we shall explore the implications of different values of the fiscal feedback parameter λ . We are interested in two key questions. First, can we distinguish clearly between two policy 'regimes', as suggested by the Fiscal Theory of the Price Level and the results in Leeper (1991) and Leith and Wren-Lewis (2000)? If we can, how does welfare and optimal monetary policy compare between regimes? Second, what is the optimal degree of fiscal feedback on debt, and what are the implications for welfare and monetary policy of departing from this optimum?

4.2 The Solution

In order to solve the model we use the method of Lagrange multipliers. Unlike the case of joint optimization in Section 3, an analysis of first order conditions is crucial to explain certain dynamic properties of the solution.

The central bank chooses the nominal interest rate to minimize social loss (21) subject to the evolution of the economy. It is instructive to simplify the dynamic system (15)-(19) that describes the evolution of out-of-steady-state economy, as observed by the monetary policymaker. We substitute equations (17) and (23) into (16), (15) and (18), leaving only three dynamic equations for c_s , π_s and \tilde{b}_s :

$$c_{s} = \mathcal{E}_{t}c_{s+1} - \sigma(i_{s} - \mathcal{E}_{t}\pi_{s+1}),$$

$$\pi_{s} = \beta \mathcal{E}_{t}\pi_{s+1} + \kappa \left(\frac{1}{\sigma} + \frac{\rho}{\psi}\right)c_{s} - \kappa \frac{\lambda}{\psi}\tilde{b}_{s} + \kappa \hat{\eta}_{s},$$

$$\tilde{b}_{s+1} = \chi i_{s} + \frac{1}{\beta} \left((1 - (1 - \tau)\lambda)\tilde{b}_{s} - \chi \pi_{s} - \tau \rho c_{s} \right) + \hat{\delta}_{s}.$$

Let the period at which optimization is taking place be period t. We construct the Lagrangian

$$\mathcal{L} = \mathcal{E}_t \sum_{s=t}^{\infty} H_s,$$

where each term H_s has the following form

$$H_{s} = \frac{1}{2}\beta^{s-t} \left[a_{\pi}\pi_{s}^{2} + a_{c}c_{s}^{2} + a_{g}\left(\frac{\lambda}{(1-\rho)}\right)^{2}\tilde{b}_{s}^{2} + a_{y}\left(-\lambda\tilde{b}_{s}+\rho c_{s}\right)^{2} \right] +\beta^{s-t}L_{s+1}^{c}\left(\mathcal{E}_{t}c_{s+1} - \sigma(i_{s}-\mathcal{E}_{t}\pi_{s+1}) - c_{s}\right) +\beta^{s-t}L_{s+1}^{\pi}\left(\beta\mathcal{E}_{t}\pi_{s+1} + \kappa\left(\frac{1}{\sigma}+\frac{\rho}{\psi}\right)c_{s} - \kappa\frac{\lambda}{\psi}\tilde{b}_{s} + \kappa\hat{\eta}_{s} - \pi_{s}\right) +\beta^{s-t}L_{s+1}^{b}\left(\chi i_{s} + \frac{1}{\beta}\left((1-(1-\tau)\lambda)\tilde{b}_{s} - \chi\pi_{s} - \tau\rho c_{s}\right) + \hat{\delta}_{s} - \tilde{b}_{s+1}\right).$$

In order to minimize the loss function in (21), we differentiate the Lagrangian, \mathcal{L} , with respect to L^c , L^{π} , L^b , π , c, b and i. The first order conditions for optimality are:

$$\frac{\partial \mathcal{L}}{\partial \pi_s} = 0 = \beta^{s-t} a_\pi \pi_s + \sigma \beta^{s-t-1} L_s^c + \beta^{s-t+1} L_s^\pi - \beta^{s-t} L_{s+1}^\pi - \beta^{s-t-1} \chi L_{s+1}^b, \tag{24}$$

$$\frac{\partial \mathcal{L}}{\partial c_s} = 0 = \beta^{s-t} a_c c_s + \beta^{s-t} a_y \rho \left(-\lambda \tilde{b}_s + \rho c_s \right) + \beta^{s-t-1} L_s^c - \beta^{s-t} L_{s+1}^c$$

$$+ \kappa \left(\frac{1}{2} + \frac{\rho}{2} \right) \beta^{s-t} I^{\pi} = -\frac{\tau \rho}{2} \beta^{s-t} I^b$$
(25)

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} a_{g} \left(\frac{\lambda}{(1-\rho)}\right)^{2} \tilde{b}_{s} - \lambda \beta^{s-t} a_{y} \left(-\lambda \tilde{b}_{s} + \rho c_{s}\right)$$

$$-\beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial \tilde{b}_{s}} = 0 = \beta^{s-t} \kappa \frac{\lambda}{\psi} \chi L_{s+1}^{\pi} + (1 - (1-\tau)\lambda) \beta^{s-t-1} L_{s+1}^{b} - \beta^{s-t-1} L_{s}^{b},$$

$$\frac{\partial \mathcal{L}}{\partial i_s} = 0 = -\sigma\beta^{s-t}L_{s+1}^c + \beta^{s-t}\chi L_{s+1}^b, \tag{27}$$

$$\frac{\partial \mathcal{L}}{\partial L_{s+1}^c} = 0 = \mathcal{E}_t c_{s+1} - \sigma (i_s - \pi_{s+1}) - c_s,$$
(28)

$$\frac{\partial \mathcal{L}}{\partial L_{s+1}^{\pi}} = 0 = \beta \mathcal{E}_t \pi_{s+1} + \kappa \left(\frac{1}{\sigma} + \frac{\rho}{\psi}\right) c_s - \kappa \frac{\lambda}{\psi} \tilde{b}_s + \kappa \hat{\eta}_s - \pi_s, \tag{29}$$

$$\frac{\partial \mathcal{L}}{\partial L_{s+1}^b} = 0 = \chi i_s + \frac{1}{\beta} \left(\left(1 - (1 - \tau)\lambda \right) \tilde{b}_s - \chi \pi_s - \tau \rho c_s \right) + \hat{\delta}_s - \tilde{b}_{s+1}, \tag{30}$$

along with initial conditions $L_t^c = L_t^{\pi} = 0, \tilde{b}_t = \bar{b}$ (see Currie and Levine (1993)).⁸

The linear difference system (24)-(30) is closed with a dynamic process for the two exogenous shocks $\hat{\eta}_{st}$ and $\hat{\delta}_t$ and this makes it to be of ninth order. It should be solved subject to nine boundary conditions. We know five initial values: (i) initial values for predetermined endogenous variables (debt in our case); (ii) Pontryagin's maximum principle requires setting to zero initial conditions for those Lagrange multipliers which are associated with dynamic constraints on nonpredetermined variables (see e.g. Currie and Levine (1993)), L^{π} and L^c in our case; (iii) and initial values of shocks $\hat{\eta}$ and $\hat{\delta}$. We need to define an appropriate transversality conditions to close the system. Welfare will be finite if variables grow slower than at a rate of $1/\sqrt{\beta}$, $\lim_{t\to\infty} \beta^{t/2} x_t = 0$ for any variable x_t . Moreover, by imposing this transversality condition we guarantee that this solution to the system (24)-(30) is unique if it exists.⁹

The system (24)-(30) plus dynamic processes for the shocks can be written in a matrix form

⁸In all the analysis below we take $\bar{b} = 0$ i.e. we assume we start from a position in which debt is at its steady state.

⁹Solutions to the dynamic system (24)-(30) would always exist if we imposed $\lim_{t\to\infty} \beta^{t/2} x_t < \infty$. However, in this case there can be multiple solutions with the asymptotic growth rate $1/\sqrt{\beta}$. Such solutions would not lead to finite welfare, but would still solve the system (24)-(30).

 $\Omega z_{t+1} = \Psi z_t$ where $z_t = (\hat{\eta}_t, \hat{\nu}_t, \tilde{b}_t, \nu_t^{\pi}, \nu_t^c, i_s, \nu_t^b, \pi_t, c_t)'$, and $\nu_t^k = \beta^{s-t} L_t^k$, $k \in \{\pi, b, c\}$. This linear system has nine generalized eigenvalues¹⁰, which are functions of the fiscal feedback parameter λ . The dynamics of the system can be fully described in terms of these eigenvalues. Moreover, the speed of convergence of the economy towards the steady state is determined by the largest eigenvalue of this system among those that are less than $1/\sqrt{\beta}$. This implies that system (24)-(30) can have solutions that exhibit a (moderate) explosion. We now examine if this is indeed the case.¹¹

To do this, it is instructive to calculate values of λ that can generate an eigenvalue $\Phi(\lambda) = 1$. We show in WP that there are two solutions to the problem where some $\Phi(\lambda) = 1$ in the area $0 \le \lambda < \infty$. They are $\lambda = 0$ and $\lambda = \lambda^*$ where

$$\lambda^* = \frac{1 - \beta}{1 - \frac{\tau\psi}{\psi + \sigma\rho}}.\tag{31}$$

(The economic interpretation of λ^* is discussed in the next section.) More specifically, at these two points for λ we have the following structure of the generalized eigenvalues of system (24)-(30).

1. If $\lambda = 0$ then the nine generalized eigenvalues to this problem are:

$$\begin{split} \Phi_1 &= \Phi_2 = \Phi_3 = 0, \\ \Phi_4 &= \frac{1}{2} \left(\frac{\left((\beta+1) + \frac{\kappa\epsilon}{\rho} \left(\frac{1}{\sigma} + \frac{\rho}{\psi} \right) \right)}{\beta} - \sqrt{\left(\frac{\left((\beta+1) + \frac{\kappa\epsilon}{\rho} \left(\frac{1}{\sigma} + \frac{\rho}{\psi} \right) \right)}{\beta} \right)^2 - \frac{4}{\beta}} \right) < 1, \\ \Phi_5 &= 1, \ \Phi_6 = \frac{1}{\beta}, \\ \Phi_7 &= \frac{1}{2} \left(\frac{\left((\beta+1) + \frac{\kappa\epsilon}{\rho} \left(\frac{1}{\sigma} + \frac{\rho}{\psi} \right) \right)}{\beta} + \sqrt{\left(\frac{\left((\beta+1) + \frac{\kappa\epsilon}{\rho} \left(\frac{1}{\sigma} + \frac{\rho}{\psi} \right) \right)}{\beta} \right)^2 - \frac{4}{\beta}} \right) > \frac{1}{\beta}, \\ \Phi_8 &= \Phi_9 = \infty > \frac{1}{\beta}. \end{split}$$

There are five generalized eigenvalues that are strictly less than $1/\sqrt{\beta}$, and the biggest of them is equal to one.

¹⁰Since Ω is singular, we deal with *generalised* eigenvalues, i.e. solutions Φ of the equation det $(\Psi - \Phi \Omega) = 0$. See Klein (2000) on solutions of such systems. In what follows we will often omit the word 'generalised', but we keep the meaning.

¹¹Note that we work with a linearised model and an infinite time horizon, so equations only remain valid in a neighbourhood of the steady state. Although *at some point* a moderate explosion would exceed these boundaries, the stability analysis is unaffected.

2. If $\lambda = \lambda^*$, then the eigenvalues are

$$\begin{split} \Phi_1 &= \Phi_2 = \Phi_3 = 0, \\ \Phi_4 &= \text{ solution of quadratic equation } < 1, \\ \Phi_5 &= 1, \ \Phi_6 = \frac{1}{\beta}, \\ \Phi_7 &= \text{ solution of quadratic equation } > \frac{1}{\beta}, \\ \Phi_8 &= \Phi_9 = \infty > \frac{1}{\beta}. \end{split}$$

There are five generalized eigenvalues that are strictly less than $1/\sqrt{\beta}$, and the biggest of them is equal to one.

These two values of fiscal feedback, 0 and λ^* , split the set of possible feedbacks into the two open sets: $0 < \lambda < \lambda^*$ and $\lambda > \lambda^*$ We can also show numerically that for a relatively wide range of $0 \le \lambda \le \Lambda$, $\Lambda \gg \lambda^*$, the following holds:

(i) some eigenvalues remain either smaller than one or greater than $\frac{1}{\beta}$ for any $0 \le \lambda \le \Lambda$: $\|\Phi_8(\lambda)\| = \|\Phi_9(\lambda)\| = \infty$, $\|\Phi_1(\lambda)\| = \|\Phi_2(\lambda)\| = \|\Phi_3(\lambda)\| = 0$, $\|\Phi_4(\lambda)\| < 1$, $\|\Phi_5(\lambda)\| > 1/\beta$. None of these eigenvalues is the 'biggest stable eigenvalue' and so none of them will determine the rate of convergence of the economy to the steady state following a shock.

(ii) Eigenvalues $\Phi_5(\lambda)$, $\Phi_6(\lambda)$ behave in the following way. First of all, they do not intersect: $\Phi_5(\lambda) < 1/\sqrt{\beta}$, and $\Phi_5(\lambda) > 1/\sqrt{\beta}$.¹² If $0 \le \lambda \le \lambda^*$, $\Phi_5(\lambda)$ increases from $\Phi_5(0) = 1$ up until it almost reaches $1/\sqrt{\beta}$ and then decreases to $\Phi_5(\lambda^*) = 1.\Phi_6(\lambda)$ decreases from $\Phi_6(0) = 1$ down until it almost reaches $1/\sqrt{\beta}$ and then increases back to $\Phi_6(\lambda^*) = 1/\beta$. For $\lambda > \lambda^*$, $\Phi_5(\lambda) < 1$ and $\Phi_6(\lambda) > 1/\beta$. $\Phi_5(\lambda)$ has a key role in determining the dynamic properties of the economy.

This dependence of eigenvalues on the value of fiscal feedback λ is shown schematically in Panel I in Figure 2, where eigenvalues are plotted against λ . Depending on the value of λ we can distinguish three cases.

1. Strong fiscal feedback when $\lambda \ge \lambda^*$. When $\lambda > \lambda^*$ we have five eigenvalues which are strictly less than one, and four explosive eigenvalues (which are strictly greater than $1/\beta > 1/\sqrt{\beta}$). Given five initial conditions and transversality conditions we obtain a unique solution. When t increases, all economic variables, \tilde{b}_t, c_t, π_t , once disturbed, necessarily converge to their steady state values. We say that in this case the solution and the steady state it converges

¹²We show this analytically in WP.

to are asymptotically stable. When $\lambda = \lambda^*$ then $\Phi_5 = 1$ and $\Phi_6 = 1/\beta$. The sixth eigenvalue is ruled out by the transversality conditions but Φ_5 is accepted by them. The optimal monetary policy thus generates unit-root dynamics of economic variables in response to shocks. We say that in this case the solution, and the steady state it does not diverge from, are just stable.

- 2. Zero fiscal feedback when $\lambda = 0$. If $\lambda = 0$ we have $\Phi_5 = 1$, $\Phi_6 = 1/\beta$. Again, transversality conditions classify Φ_5 as a just stable eigenvalue. We thus obtain a unique solution. We can check that in this case neither of the economic variables b_t, c_t, π_t nor instrument i_t will exhibit unit-root behavior, as the Lagrange multipliers have unit root dynamics.
- 3. Weak fiscal feedback when $0 < \lambda < \lambda^*$. For the intermediate range of parameter λ , $0 < \lambda < \lambda^*$, there are five eigenvalues that are less than $1/\sqrt{\beta}$, but one of them is greater than one. The model thus exhibits explosive behavior. This explosive behavior is modest, as variables grow at an asymptotic rate that is slower than the steady state rate of interest, $1/\beta$. The implied loss is finite.

These results suggest that we can define three different *regimes*, determined by the two threshold values of parameter λ . Two regimes, the zero fiscal feedback regime and the strong fiscal feedback regime, are regimes with stable solutions. The third regime, the weak fiscal feedback regime, generates a moderately explosive solution that delivers finite social loss. Although such behavior is theoretically possible, we are not aware of another example in the literature where it would be *optimal* for monetary policy to support moderate explosion. Although this distinction is made on the basis of dynamic stability properties, we shall see in the next section that there is a substantial difference among these regimes in the implied economic behavior of the policymaker.

4.3 Optimal Monetary Policy and Welfare

Panel II in Figure 2 presents the values of some key magnitudes as we change the degree of fiscal feedback λ . The first two columns are identical except for the scale of λ : the first column focuses on small positive values of fiscal feedback, whereas the second column gives results for a much broader range.

The top two rows report monetary policy responses to the cost-push shock and debt. It is well known (see Klein (2000) or WP) that the unique solution for the optimal interest rate reaction function in linear-quadratic models can be presented in the form of a linear relationship:

$$i_t = \theta_\eta \hat{\eta}_t + \theta_\zeta \hat{\zeta}_t + \theta_b \tilde{b}_t + \vartheta_\pi L_t^\pi + \vartheta_c L_t^c, \tag{32}$$

with feedback coefficients θ on predetermined states and predetermined Lagrange multipliers. We plot two parameters from the implied reaction function for monetary policy (32): the feedback coefficient on the mark-up shock θ_{η} and on debt θ_b .¹³. Solid lines indicate solutions that are asymptotically stable, while dotted lines are solutions that involve mild explosions (see the discussion in the previous section).

The bottom row of Panel II plots the social welfare loss expressed as a percent reduction in steady-state consumption. We discuss below the size of these welfare losses, and compare them to the results in the relevant literature.

4.4 Three policy regimes

Figure 2 and the analysis in Section 4.2 suggest that there are two regimes in which all processes are either asymptotically stable or unit root, and one in between which exhibits moderate explosive behavior. We discuss them in turn.

The first regime occurs when $\lambda = 0$. Here we have no feedback from debt to fiscal variables, so we might suppose that debt in this model would be unstable. However, the results show that monetary policy ensures the *asymptotic* stability of economic variables. It achieves this in two ways, both shown in Panel II in Figure 2. First, θ_b ensures that any positive movement of debt leads to a large fall in interest rates, which leads to correction through the government's budget constraint. Second, the reaction to a positive cost-push shock is also to reduce interest rates.

The negative feedback on debt is not surprising, given inaction by the fiscal authorities. The negative reaction to the cost-push shock is more interesting, and it raises the question of how monetary policy stabilizes inflation in this case. To understand what is going on, Panel I in Figure 3 plots impulse responses to the cost-push shock for two cases, both of which set $\lambda = 0$: optimal monetary policy (solid line), and for fixed nominal interest rates, which is a standard example of a passive monetary policy in the literature. The cost push shock raises inflation, and when interest rates are fixed this reduces debt. However, when inflation falls back, it returns to a small negative number, and from there gradually converges to zero. This reduction in inflation slowly increases debt, and allows it to return to its initial level. When monetary policy is optimal, then in the first period there is a large reduction in interest rates, as Panel II in Figure 2 also shows, but this is followed by an increase in interest rates. The key to why this path is optimal is that both consumption and inflation are forward looking. Higher future interest rates largely offset

¹³The other ϑ parameters are less informative. ϑ_L represent the integral control part of the reaction function and therefore feedback on slow moving variables.

the impact of the immediate decline in interest rates on first period consumption, and thereafter consumption is below base. Inflation depends on current and future consumption, so inflation is lower in all periods as a result of this behavior in consumption. Of course debt depends on its own past value through the budget constraint. Thus, by cutting interest rates initially and raising them subsequently, monetary policy is able to both stabilize debt and moderate the initial increase in inflation.¹⁴

In contrast, where $\lambda \geq \lambda^*$, the reaction to the cost-push shock is generally to raise interest rates. The system is asymptotically stable for $\lambda > \lambda^*$ and exhibits a unit-root behavior for $\lambda = \lambda^*$. The contrast between behavior in the two regimes is illustrated in Panel II in Figure 3, which plots the impulse response for key economic variables following a cost push shock for two values of λ : $\lambda = 0$ as a dashed line, and $\lambda = \overline{\lambda} > \lambda^*$ as a solid line, where $\overline{\lambda}$ is only marginally higher than λ^* so the solution is asymptotically stable.

As we would expect from simply considering the government's budget constraint, the value λ^* that produces unit-root behavior in debt will be a function of the steady state real interest rate. However it is also a function of the tax rate, as the reduced form debt equation in Section 4.2 shows.¹⁵ A reduction in government spending will reduce debt directly, but it also reduces income and therefore income taxes, which raises debt. Thus λ^* has to be greater than the steady state real interest rate to just stabilize debt.

These two regimes are separated by a region $0 < \lambda < \lambda^*$ where the system is not asymptotically stable, but has a 'moderately' explosive solution. In this case fully optimal monetary policy chooses not to ensure the convergence of the economy back to the steady state. As we have seen above, there is a clear conflict for monetary policy in both stabilizing debt and stabilizing inflation when fiscal feedback fails to stabilize debt. As a result, monetary policy pushes close to the boundary one of these objectives, which is to allow debt to mildly explode in a manner that keeps the welfare loss finite and allows policy to reduce the initial impact on inflation.

There is a clear parallel between our results using optimal monetary policy and the active and passive regimes described by Leeper (1991) and Leith and Wren-Lewis (2000). In their case a passive monetary policy is defined as a negative response of real interest rates when inflation is

¹⁴Although optimal policy produces larger deviations from the inflation target after the initial period than under the fixed interest rate policy, the convexity of the welfare function implies that the impact of this on welfare is more than offset by the impact of the reduction in inflation in the initial period.

¹⁵Leith and Wren-Lewis (2000) use a determinate condition (which is a necessary but not sufficient condition for a determinate or stable solution) to calculate analytically a value of fiscal feedback of a similar order of magnitude that divides their two policy regimes. They can do this because they use a Taylor rule to describe monetary policy, rather than calculate optimum monetary policy.

above target in a Taylor rule, whereas in our case it corresponds to a substantial fall in real interest rates following a positive cost-push shock. When there is no fiscal feedback, fiscal instruments do nothing to prevent a debt interest spiral. To avoid an explosive solution for debt, monetary rather than fiscal policy must stabilize the government's debt stock, along the lines discussed above. However, unlike the literature cited above, our alternative regime does not always involve an active monetary policy, a point which we return to when we look at alternative steady state values of debt.

4.5 Fiscal feedback and welfare

As Panel II in Figure 2 shows, there is a non-trivial difference in the levels of welfare attained in the two asymptotically stable regimes, which at its greatest is around 0.25% of steady state consumption. While the papers cited above, and the Fiscal Theory of the Price Level more generally, have shown that a lack of fiscal feedback does not necessary lead to model instability, it is clear from our results that weak or zero fiscal feedback seriously damages the ability of monetary policy to reduce the social costs of macroeconomic shocks. These results are in contrast with those obtained by Schmitt-Grohe and Uribe (2007) where the difference does not exceed 0.05% of steady state consumption. This contrast reflects our inclusion of cost-push shocks, which in models of this kind are the most problematic for monetary policy. Taste/technology shocks are much less important because, with optimal monetary policy, they only influence our economy through their impact on debt, as we showed in Section 2.5, and the gain from eliminating these effects is small.

The optimal value of λ (which we denote as $\overline{\lambda}$, and which was used to plot dashed lines in Panel II in Figure 3) is very close to the lowest possible value that sustains this regime, λ^* . As we noted above, at $\lambda = \lambda^*$, one of the system's eigenvalues is exactly unity, and this corresponds with a unit root process for debt. At the optimal value of lambda ($\lambda = \overline{\lambda}$), therefore, debt is *almost* a unit root process, but will eventually return to its original steady state value. We saw in Section 3 that joint monetary-fiscal optimization would imply a unit root process for debt. However, we noted from Figure 1 that under fully optimal fiscal policy the debt implications of the shock are not completely accommodated: there is an attempt in the first period to reduce spending and thereby moderate the eventual increase in debt. This initial path for government spending cannot be replicated under our simple feedback rule, because spending is tied to debt. Although this short term difference is small in quantitative terms, it helps explain why the optimal level of fiscal feedback ($\overline{\lambda}$) is very slightly above that required for a unit root process. The optimal value of fiscal feedback, although it does not produce a unit root process, is extremely close to one: debt is substantially above its original level even after 500 years. (This is shown as p = 0.0 case in Figure 6, Panel II.) The value $\lambda = \overline{\lambda}$ implies that for every \$100 that debt is above its steady state level, government spending is reduced by \$1.25 a quarter.

For $\lambda > \overline{\lambda}$, Panel II in Figure 2 shows that the welfare loss steadily increases, although even when adjustment becomes large (a value of $\lambda = 0.3$ implies that government spending falls by \$30 each quarter for every \$100 that debt is above steady state), the loss is never as great as in the case of $\lambda = 0$. However, the increase in loss does demonstrate the macroeconomic costs involved in attempting to correct debt too quickly when government spending is the fiscal instrument.¹⁶ While a policy that set λ a little above $\overline{\lambda}$ would have little cost, setting a much larger value for λ would incur significant costs.

Increasing λ beyond its optimal value does have a noticeable impact on optimal monetary policy: the response of interest rates to the cost push shock initially falls as λ increases, and becomes negative for a time. One reason for this is as follows. For large λ , fiscal policy helps stabilize the impact of a cost push shock. The shock raises debt (see above), which with large λ implies a substantial decline in government spending. This deflates the economy, implying less of a need for real interest rates to rise. However, this form of feedback is less efficient at demand stabilization than monetary policy, as the values for welfare show. Although both fiscal and monetary policy act directly on demand (through public and private consumption respectively), fiscal policy only acts when debt changes, whereas optimal monetary policy can respond directly to inflationary shocks, and is therefore more efficient. These results also suggest that we cannot characterize this policy regime as *always* involving an active monetary policy, a point that is reinforced (and explained) when we look at higher initial debt levels. However, monetary policy is always considerably more active than when there is minimal fiscal feedback.

Finally, we compare welfare under joint monetary-fiscal optimization with welfare when fiscal feedback is optimal. We argued above that, given current institutional arrangements, fiscal feedback represents a more realistic view of fiscal policy setting than a fully optimal fiscal policy, but it is interesting to note what the costs of this are. The difference between welfare in the two policies can be observed by comparing Table 1, (when government spending is the only instrument) with welfare when fiscal feedback is at the optimal $\bar{\lambda}$. This amounts to only 0.002% of steady state consumption. In this case, therefore, there is only a small cost in restricting fiscal policy to respond to debt alone.

¹⁶The costs of larger λ 'come from' the quadratic term in g in social welfare. If we artificially delete this term, the loss function after λ^* would be flat. However, as we noted above, even if all government spending was pure waste, g would still influence social welfare.

5 Generality of the Results

5.1 Tax rate as an instrument

Our choice of government spending rather than the income tax rate as the dependent variable in the fiscal feedback rule was essentially arbitrary. One argument in favour of using taxes rather than spending is that the latter is less flexible, and some components of spending may be effectively exogenous. If we used income taxes instead, then the fiscal feedback rule would become

$$\left(\Upsilon_t - \Upsilon_t^n\right)Y = \omega\left(B_t - B\right),\tag{33}$$

and its log-linearized version

$$\tau_t = \frac{\omega}{\tau} \tilde{b}_t. \tag{34}$$

It can be shown (see WP) that if we put $\omega \neq 0$ but impose $\lambda = 0$ then the system of first order conditions is structurally very similar to the one where we used g_t as an instrument. We can easily obtain similar analytical results: (i) if $\omega = 0$ then there is an asymptotically stable regime with passive monetary policy, (ii) if $\omega \geq \omega^*$ where

$$\omega^* = \frac{(1-\beta)}{\left(1 - \frac{\tau\psi\sigma\rho}{(1-\tau)(\psi+\sigma\rho)}\right)},\tag{35}$$

then there is an asymptotically stable regime with active monetary policy and (iii) if $0 < \omega < \omega^*$ then there is a moderately explosive regime.

Panel I in Figure 4 repeats Panel II in Figure 2 for varying ω but keeping $\lambda = 0$; i.e. fiscal feedback involves income taxes, and not government spending. Although the pattern is broadly the same as in Figure 2, there are three notable differences. First, at the optimal feedback parameter, welfare is slightly better than with feedback on government spending. Second, this optimal value is just below, rather than just above, the value of feedback associated with a unit root debt process (ω^*).¹⁷ Third, as ω increases beyond ω^* , the welfare cost of the cost-push shock increases at a much more gradual rate than when government spending was the fiscal instrument. The reason for this is that government spending impacts directly on demand, whereas taxes work

¹⁷Recall that the fully optimal path of taxes following a cost-push shock involved a large initial cut, which was then reversed. Clearly a simple feedback rule on debt cannot replicate this, but feedback that is just below ω^* makes a partial attempt. In contrast, when government spending was the fiscal instrument, its optimum path overshot its long run level level, so $\bar{\lambda} > \lambda^*$ in that case.

through consumption and labour supply. The income effect will be smoothed by consumers, so strong feedback on debt will interfere less with monetary policy.

What if we used both government spending and taxes? If we vary (λ, ω) over the domain $((\lambda, \omega) \in [0, \Lambda] \times [0, \Omega], \Lambda \gg \lambda^*, \Omega \gg \omega^*)$ then we can plot the value of losses as a function of λ and ω .Panel II in Figure 4 plots level contours in the non-explosive area. The minimum loss is achieved with a mixed policy, but where the feedback on government spending is small. As a result, this policy is similar to the tax only policy discussed above.

5.2 Varying the steady state level of debt

We look at two alternative steady state levels of debt, zero and doubling the base case to 60% of GDP.¹⁸ The latter still involves levels that are well below those in many industrialized countries. However our model is not complex enough to distinguish between debt of different maturities, and so we may be overestimating the impact of changes in short term rates on debt interest payments. As a result, a conservative choice of steady state debt levels seems appropriate. The key parameters that determine the debt accumulation process are ρ, τ and χ . However, not all of them are independent. If we take the share of government spending to output $(1 - \rho)$ as given, then there is a relationship that links the steady state level of the tax rate τ with the steady state level of debt to output ratio, χ :

$$\tau = (1 - \beta)\chi + 1 - \rho. \tag{36}$$

This relationship either determines τ for given χ , or determines χ given τ . In what follows we assume that χ and ρ determine τ . The higher the level of debt the higher the steady state level of taxes. Higher taxes widen the area over which optimal monetary policy produces outcomes that are not asymptotically stable, as formulae (31) and (35) show. (See the discussion of λ^* in Section 4.3.)

If the steady state level of debt is zero ($\chi = 0$) then the linearization in Section 2 should be changed as we cannot construct \hat{B}_t . As $\chi = 0$ then for small disturbances B_t itself will be 'small', so the correct version of the linearized budget constraint is

$$\tilde{b}_{t+1} = \frac{1}{\beta} (\tilde{b}_t + (1-\rho)g_t - \tau y_t),$$
(37)

but where $\tilde{b}_t = B_t$. There are two differences between expressions (37) and (18). First, there are no first-order effects of interest rates and inflation on debt in (37) and, second, taste/productivity

¹⁸For simplicity, we assume that government spending is the fiscal instrument.

shocks do not have first-order effect on debt if B = 0. Fiscal rule (23) remains the same, but the notation \tilde{b}_t is recycled. We therefore need to solve the system (15), (16), (17), (37) and (23). Again, we can solve the system of first order conditions and obtain that there is a region of explosion when $0 < \lambda < \lambda^*$ where λ^* is given by the same formula (31) but where $\tau = 1 - \rho$ as follows from formula (36).

If $B \neq 0$ then monetary policy affects debt via two channels. The first channel is direct: a change in interest rates has a one-to-one effect on debt. The second channel is indirect: a change in interest rates influences price setting and consumption decisions, which impact on output and taxes. By putting B = 0 we eliminate the first channel but retain the ability to affect debt via the second channel.

Panel I in Figure 5 repeats Panel II in Figure 2 for the three levels of initial debt. The different policy regimes in terms of feedback parameters and welfare costs are evident in all cases, but there are some interesting differences. First, when fiscal feedback is zero, welfare losses are greatest when the steady state debt is also zero. This follows from the fact that in this case monetary policy is required to stabilize debt, but its ability to do so is severely weakened by the absence of a debt interest channel. The higher is debt, the less costly the absence of fiscal feedback is. Second, in the high debt case, the feedback parameter on the cost push shock remains negative, even when fiscal feedback becomes significant. This change in also reflected in the actual movement of interest rates following the cost-push shock, as Panel II in Figure 5 shows. Interest rates initially fall following a cost push shock, although they increase quite quickly thereafter. The consequences of a large initial stock of debt are therefore to delay the point at which interest rates rise. The reasons for this are very similar to our discussion of Figure 3 when there was no fiscal feedback. A key difference is that with no fiscal feedback monetary policy was forced to delay the increase in interest rates because it had to control debt, whereas in this case it is simply preferable to use monetary policy rather than fiscal feedback to control debt.

This result suggests that the link between optimal monetary policy and fiscal feedback on debt is rather more complex than a simple 'no feedback = passive monetary policy, significant feedback = active monetary policy' equation. If we compute a fully optimal fiscal policy when debt is high both the direct feedback of interest rates on the cost-push shock and the initial change in interest rates is negative. However, it remains the case that monetary policy is always more active/less passive when fiscal feedback is at or above its optimal level compared to when fiscal feedback is negligible. This remains true even if steady state levels are doubled yet again to over 120% of annual GDP.

5.3 Sensitivity to source of shocks

As we noted in Section 4.3, welfare losses are dominated by the impact of the cost-push shock. To what extent might our results therefore be specific to this shock? To examine this, we repeated the loss chart from the first two columns in Panel II in Figure 2 setting the cost push shock to zero. As the chart in the third column of the same panel shows, the basic characteristics of welfare as fiscal feedback varies are identical, although of course the size of the welfare losses are much smaller. The size of the welfare losses shown in this case are much more comparable to those in Schmitt-Grohe and Uribe (2007). This indicates that our basic results are robust to choice of shocks.

5.4 Blanchard-Yaari consumers and intergenerational effects

The results discussed so far assume that consumers are infinitely lived, so changes in government debt/personal wealth have no direct effect on the pattern of consumer spending over time. In this section we examine an alternative set up, where consumers have finite lives, using the framework due to Blanchard and Yaari (Blanchard (1985)). (Blanchard/Yaari consumers are also modelled in Leith and Wren-Lewis (2000), who examine issues of stability and monetary/fiscal policy interaction in a monetary union, as well as Smets and Wouters (2002) and Ganelli (2005)). With Blanchard/Yaari consumers, we now have a direct route whereby changes in government debt will influence changes in consumption, and we want to examine the extent to which the results described above continue to hold. Introducing Blanchard/Yaari consumers does, however, introduce costs in terms of complexity, which is why we do not examine them in the base case. Appendix A outlines the changes to the model when consumers have finite lives.

Panel I in Figure 6 repeats the analysis shown in Panel II in Figure 2 when government spending is the fiscal instrument. The broad pattern is the same, but there is one significant difference. The area where the economy is not asymptotically stable shifts slightly to the right. (We choose a deliberately high value for the probability of death for this figure so that this point is clear.) This is consistent with results in Leith and Wren-Lewis (2000), where the critical value of λ derived from the determinate stability condition is a positive function of p.¹⁹ The economic

¹⁹Our results go beyond those in Leith and Wren-Lewis (2000), who also consider Blanchard–Yaari consumers, in three respects. First, we show for negligible fiscal feedback that the optimal monetary policy is still passive (it responds negatively to inflation) even though it can also feedback directly from debt. Second, we show that the optimal monetary policy is strongly passive: the negative feedback on the cost push shock and inflation is very large. (This result is hinted at, but not established, in Leith and Wren-Lewis (2000).) Third, Panel I in Figure 6 shows that this passive monetary policy, while it stabilizes debt, has a clear welfare cost compared to the alternative regime with significant fiscal feedback.

reason for this is as follows. A cost push shock with an active monetary policy raises debt, and this has a positive impact on demand through consumption with Blanchard Yaari consumers. As a result, monetary policy will generate a larger increase in interest rates, which in turn requires a larger decrease in government spending to prevent a debt interest spiral. In fact, there is a natural neutrality result here. The net impact of debt on demand combines the positive wealth effect from Blanchard–Yaari consumers with the negative effect operating through fiscal feedback. It seems logical that if the former increases (because of larger p), then optimal λ should rise in a corresponding way, thereby neutralizing the overall impact of debt on demand.

We also compute the joint monetary-fiscal optimal policy when we have Blanchard Yaari consumers. Recall that with infinitely lived consumers, this policy implied a unit root process for debt, a result that is consistent with findings in Schmitt-Grohe and Uribe (2004) and Benigno and Woodford (2004). However, there has until now been no equivalent analysis in a model where consumers have finite lives and there are no bequests. We find that the random walk result does not hold in this case. Joint monetary-fiscal optimization produces a system where one of the eigenvalues is very close to one, but not equal to one. The reason for this is as follows. In a model with Blanchard Yaari consumers, the steady state real interest rate is no longer always equal to the rate of time preference, but instead is increasing in the steady state level of debt. A standard result from consumption smoothing is that if the real rate of interest differs from the rate of time preference we get 'tilting', and the same applies in this case to the path of public consumption chosen when the policy maker optimizes. This makes a pure random walk outcome suboptimal. However, as Panel II in Figure 6 shows, for realistic values of the probability of death the behavior of debt, both for a fully optimal policy and for optimal fiscal feedback, are very close to a unit root process, with less than half of any debt in excess of steady state eliminated after 250 years.

In this analysis, we have continued to assume that the monetary policy maker maximizes social welfare, which includes the welfare of unborn generations. (For a discussion of how this is done, see Appendix A.) However, this masks the potential for intergenerational conflict when fiscal feedback is modest. The right hand picture in Panel III in Figure 6 plots the difference between the *per period* social welfare loss for two values of λ : specifically $H_s(\lambda = 0.05) - H_s(\lambda = \lambda^*)$, where H_s is the per period loss at time s. In the first few periods the loss of feeding back on debt with large feedback outweighs the loss of the value of feedback that produces a unit root process in debt. After a number of periods, however, the (constant) loss from having unit root dynamics outweighs the loss of having strong feedback; in the latter case the economy is brought back to the steady state, so the loss will become zero. It is clear that from the point of view of future

generations, a level of fiscal feedback that comes close to a unit root process for debt (i.e. the socially optimal $\bar{\lambda}$) is *not* preferred to one where fiscal feedback is more rapid. By implication, current generations will prefer a level of fiscal feedback that is less rapid than $\bar{\lambda}$.

We can confirm this by assuming that the monetary authority is 'captured' by currently living generations, and so maximizes an objective function which discounts per period social welfare at a rate equal to $\beta/(1+p)$, rather than β . In this case the monetary authority can allow debt to explode at a rate less than $\sqrt{(1+p)/\beta}$. An interesting result is that the optimal value (from the monetary policymaker's point of view) of fiscal feedback in this case becomes very small, at almost zero. In effect, current generations are able to disregard the debt problem completely, because it only affects future generations, and therefore prefer a value of fiscal feedback that allows them to maximize their impact on inflation. At this new optimal level of fiscal feedback debt explodes at a rate that is greater than $1/\sqrt{\beta}$, but which is still less than $\sqrt{(1+p)/\beta}$. We plot the policymaker's loss as a dashed line in the second picture in the Panel, indicating by a dotted line the area of explosion. Social welfare, plotted as a solid line, is only finite when the rate of explosion is smaller than $1/\sqrt{\beta}$, so the social loss becomes infinite in a neighborhood of the captured policymaker's best choice of fiscal feedback.

6 Conclusion

We have examined the impact of different degrees of fiscal feedback on debt in an economy with nominal inertia where monetary policy is optimal. Consumers are either infinitely lived, or of the Blanchard Yaari type. Our focus is on the extent to which different speeds of fiscal feedback on debt enhance or detract from the ability of the monetary authorities to stabilize output and inflation.

We use a welfare function derived from utility, and calculate joint fiscal-monetary optimization (i.e. a fully optimal fiscal policy with no constraints) as a benchmark. If consumers are infinitely lived we find that debt follows a unit root process, as found in Schmitt-Grohe and Uribe (2004) and Benigno and Woodford (2004). However, if consumers are of the Blanchard Yaari type, joint fiscal-monetary optimization no longer involves an exact unit root process, although it is close to it. If we then restrict fiscal policy to follow a simple feedback on debt, a formulation which seems closer to current institutional practice, we find the optimal level of fiscal feedback to be small. With this optimal degree of fiscal feedback, the behavior of debt is very close to a unit root process following shocks.

At low or moderate levels of initial debt we directly infer that optimal monetary policy is

active, in the sense that real interest rates initially rise following an increase in inflation, both for fully optimal fiscal policy and for the optimal level of fiscal feedback. In addition, we find that the costs of restricting fiscal policy to only respond to debt are small compared to a fully optimal fiscal policy, if government spending is the fiscal instrument. We also show that fiscal feedback using taxes rather than government spending is slightly preferable in welfare terms. If the initial debt stock is large so that changes in interest rates have a large impact on the government's budget constraint, then monetary policy may no longer be active under optimal fiscal feedback or joint fiscal-monetary optimization.

There is a discontinuity in the behavior of monetary policy and welfare either side of this optimal level of fiscal feedback. As the extent of fiscal feedback increases beyond the optimal level, optimal monetary policy becomes less active because fiscal feedback also tends to deflate inflationary shocks. However this fiscal stabilization is less efficient than monetary policy, and so welfare declines. In contrast, if fiscal feedback falls below the optimal level, then optimal monetary policy initially permits solutions that are mildly explosive, so that they are not asymptotically stable. In addition monetary policy becomes much more passive in nature, in the sense that interest rates initially fall sharply despite higher inflation. When fiscal feedback becomes zero optimal monetary policy remains passive, but the economy is asymptotically stable. This policy regime has strong similarities to the Fiscal Theory of the Price Level. We show that while this passive monetary policy may succeed in controlling debt, it leads to a sharp deterioration in welfare.

A The Model with Blanchard Yaari consumers

We briefly outline changes to the model and the welfare metric in this section: further details are given in WP and in Kirsanova et al. (2007). We need to make a number of changes to our model, described by equations (15)–(19). First, as consumers have a constant probability of death, p, the discount factor in formula (1) becomes $\beta/(1+p)$. Second, in the household budget constraint (2), the discount factor takes account of mortality, $\mathcal{E}_t(Q_{t,t+1}) = \frac{1}{(1+i_t)(1+p)}$. Third, these modifications and the fact that we now have an infinite number of living cohorts at each moment of time, results in a new system for aggregate variables. The first order conditions for individual consumption, and then aggregation across all generations, yield a pair of equations for aggregate consumption and for the average propensity to consume, instead of the single Euler equation (7):

$$\hat{C}_t = [\beta(1+i)]^{-\sigma} (\mathcal{E}_t \hat{C}_{t+1} + \frac{p\rho}{\Phi\theta} (\mathcal{E}_t \hat{A}_{t+1} - \mathcal{E}_t \hat{\pi}_{t+1} - \mathcal{E}_t \hat{\Phi}_{t+1})) - \sigma(\hat{\imath}_t - \mathcal{E}_t \hat{\pi}_{t+1}),$$
(38)

$$\frac{(1+p)(1+i)}{\beta^{\sigma}(1+i)^{\sigma}}\hat{\Phi}_{t} = \mathcal{E}_{t}\hat{\Phi}_{t+1} - (1-\sigma)(\hat{\imath}_{t} - \mathcal{E}_{t}\hat{\pi}_{t+1}),$$
(39)

where $1/\Phi_t$ is average propensity to consume out of total resources, resources which consist of nominal financial wealth and human wealth. Equations (38) and (39) can be written in terms of gap variables, and they should replace equation (15).

To evaluate gains and losses we need a welfare metric. In the Blanchard-Yaari case, unlike in the infinitely-lived case, there is no obvious choice. Ideally total welfare should be evaluated using a social welfare function that aggregates across generations and weights the utility of every generation, but straightforward aggregating of the utilities of unborn generations is not feasible for computational reasons. One way to overcome this difficulty is to suggest that the government uses a weighting scheme that makes the aggregate welfare of overlapping generations equivalent to the welfare of one infinitely long lived generation of consumers. A similar strategy was also adopted by Calvo and Obstfeld (1988). We therefore use formulae (21) to obtain our results.

References

Auerbach, A. J. (2002). Is there a Role for Discretionary Policy? NBER Working Paper 9306.

- Barro, R. J. (1979). On the Determination of Public Debt. *Journal of Political Economy 87*, 940–971.
- Beetsma, R. and H. Jensen (2004). Mark-up Fluctuations and Fiscal Policy Stabilization in a Monetary Union. Journal of Macroeconomics 26, 357–376.
- Benigno, P. and M. Woodford (2004). Optimal Monetary and Fiscal Policy: A Linear-Quadratic Approach. In NBER Macroeconomics Annual 2003, pp. 271–333.
- Blanchard, O. (1985). Debt, Deficits, and Finite Horizons. Journal of Political Economy 93(2), 223–247.
- Buiter, W. (2002). The fiscal theory of the price level: A critique. *Economic Journal 112*, 459–480.
- Calvo, G. and M. Obstfeld (1988). Optimal Time-Consistent Fiscal Policy with Finite Lifetimes. Econometrica 56(2), 411–432.
- Canzoneri, M., R. Cumby, B. Diba, and D. Lopez-Salido (2006). Monetary and Fiscal Policy Coordination when Bonds Prvide Transaction Services. Mimeo, Georgetown University.

- Canzoneri, M., R. Cumby, B. Diba, and O. Mykhaylova (2006). New Keynesian Explanations of Cyclical Movements in Aggregate Inflation and Regional Inflation Differentials. Open Economies Review 17(1), 27–55.
- Currie, D. and P. Levine (1993). Rules, Reputation and Macroeconomic Policy Coordination. Cambridge: Cambridge University Press.
- Dixit, A. and J. Stiglitz (1977). Monopolistic Competition and Optimum product Diversity. *American Economic Review* 67, 297–308.
- Favero, C. and T. Monacelli (2005). Fiscal Policy Rules and Regime (In)Stability: Evidence from the U.S. Mimeo, University of Bocconi.
- Ganelli, G. (2005). The new open economy macroeconomics of government debt. *Journal of International Economics* 65, 167–184.
- Ireland, P. N. (2004). Technology Shocks in the New Keynesian Model. Review of Economics and Statistics 86(4), 923–936.
- Kirsanova, T., M. Satchi, D. Vines, and S. Wren-Lewis (2007). Optimal Fiscal Policy Rules in a Monetary Union. Journal of Money, Credit and Banking 39(7), 1759–1784.
- Kirsanova, T. and S. Wren-Lewis (2007). Optimal fiscal feedback on debt in an economy with nominal rigidities. Working Paper, University of Exeter.
- Klein, P. (2000). Using the Generalized Schur Form to Solve a Multivariate Linear Rational Expectations Model. Journal of Economic Dynamics and Control 24 (10), 1405–1423.
- Leeper, E. M. (1991). Equilibria Under 'Active' and 'Passive' Monetary and Fiscal Policies. Journal of Monetary Economics 27, 129–147.
- Leith, C. and S. Wren-Lewis (2000). Interactions Between Monetary and Fiscal Policy Rules. The Economic Journal 110, C93–C108.
- Leith, C. and S. Wren-Lewis (2007). Fiscal Sustainability in a New Keynesian Model. University of Oxford Working Paper No. 310.
- Muscatelli, A., P. Tirelli, and C. Trecroci (2004). Fiscal and monetary policy interactions: Empirical evidence and optimal policy using a structural New-Keynesian model. *Journal of Macro*economics 26, 257–280.

- Rudebusch, G. (2002). Assessing Nominal Income Rules for Monetary Policy with Model and Data Uncertainty. *Economic Journal* 112, 402–432.
- Schmitt-Grohe, S. and M. Uribe (2004). Optimal fiscal and monetary policy under sticky prices. Journal of Economic Theory 114, 198–230.
- Schmitt-Grohe, S. and M. Uribe (2007). Optimal Simple and Implementable Monetary and Fiscal Rules. Journal of Monetary Economics 54(6), 1702–1725.
- Smets, F. and R. Wouters (2002). Openness, Imperfect Exchange Rate Pass-Through and Monetary Policy. *Journal of Monetary Economics* 49.
- Smets, F. and R. Wouters (2003). An Estimated Dynamic Stochastic General Equilibrium Model of the Euro Area. *Journal of the European Economic Association 1*.
- Söderlind, P. (1999). Solution and Estimation of RE Macromodels with Optimal Policy. European Economic Review 43, 813–823.
- Sutherland, A. (2002). A Simple Second-Order Solution Method For Dynamic General Equilibrium Models. CEPR discussion paper 3554.
- Taylor, J. B. (2000). Reassessing Discretionary Fiscal Policy. Journal of Economic Perspectives 14(3), 31–36.
- Woodford, M. (1996). Control of the Public Debt: A Requirement for Price Stability? Working Paper 5684, NBER.
- Woodford, M. (2003). Interest and Prices: Foundations of a Theory of Monetary Policy. Princeton, NJ.: Princeton University Press.



Figure 1: Responses to a unit cost-push shock under fully optimal monetary and fiscal policy using alternative fiscal instruments.



Panel I: STRUCTURE OF EIGENVALUES

Figure 2: The structure of eigenvalues, coefficients of monetary policy reaction function and social welfare as a function of fiscal feedback. Fiscal policy uses government spending as an instrument.

Panel I: ZERO FISCAL FEEDBACK



Panel II: ACTIVE AND PASSIVE MONETARY POLICY



Figure 3: Impulse responses to a unit cost-push shock, plotted for alternative feedback parameters. Fiscal policy uses government spending as an instrument.



Panel I: SOLUTION WHEN TAXES ONLY ARE USED

Figure 4: Coefficients of monetary policy reaction function and social welfare as a function of fiscal feedback. Fical policy uses taxes as an instrument in Panel I and uses taxes and spending in Panel II.

0.015 ω

0.02

0.025

0.03

0.01

0 L 0

0.005



Panel I: MODEL SOLUTION FOR DIFFERENT VALUES OF B

Panel II: IMPULSE RESPONSES FOR λ = 0.05 AND DIFFERENT VALUES OF B



Figure 5: Coefficients of monetary policy reaction function and social welfare as a function of fiscal feedback plotted for alternative values of steady state debt (Panel I), and Impulse responses to a unit cost push shock (Panel II). Fiscal policy uses spending as an instrument.



Figure 6: Panel I and Panel II illustrate the effect of Blanchard-Yaari consumers on the monetary reaction function and social welfare, and impulse responses to a unit cost-push shock respectively. Panel III illustrates the effect of policymaker's discounting on the choice of optimal fiscal feedback.